Author’s response to reviews

Title: Smoking cessation can improve quality of life among COPD patients: Translation and validation of the clinical COPD questionnaire into Greek

Authors:

George Papadopoulos (gnpapa@gmail.com)
Constantine I Vardavas (vardavas@edu.med.uoc.gr)
Maria Limperi (mlimperi@yahoo.com)
Apostolos Linardis (alinardis@ekke.gr)
George Georgoudis (gg@hol.gr)
Panagiotis Behrakis (pbehraki@hsph.harvard.edu)

Version: 5 Date: 22 February 2011

Author’s response to reviews:

Dear editorial production team,

Reference 9

is in et al, as it has more than 30 authors.

All other changes requested have been made,

Constantine Vardavas